Author's response to reviews

Title: Driving patterns in older adults with glaucoma

Authors:

Suzanne W van Landingham (swestbr3@jhmi.edu)
Chad Hochberg (chadhochberg@gmail.com)
Robert W Massof (bmassof@jhmi.edu)
Emilie Chan (emiliechan@gmail.com)
David S Friedman (David.Friedman@jhu.edu)
Pradeep Y Ramulu (pramulu1@jhmi.edu)

Version: 2 Date: 29 May 2012

Author's response to reviews:

We have made two minor changes: we have changed the wording of the title and the order of variable presentation in the tables. Apologies for not doing this before submission.